Author’s response to reviews

Title: Breast Cancer Screening Disparities among Urban immigrants: A Population-Based Study in Ontario, Canada

Authors:

Mandana Vahabi (mvahabi@ryerson.ca)
Aisha K Lofters (aisha.lofters@utoronto.ca)
Richard H Glazier (rick.glazier@ices.on.ca)

Version: 2 Date: 18 September 2014

Author’s response to reviews:

Dear Editor,

We included all the requested changes indicated in your email dated September 18. These included insertion of line numbering, addition of abstract to the manuscript, co-authors’ email addresses and competing interest section after the conclusion.